Author's response to reviews

Title: Analysis of association of MEF2C, SOST and JAG1 genes with bone mineral density in Mexican-Mestizo postmenopausal women.

Authors:

Rafael Velázquez-Cruz (rvelazquez@inmegen.gob.mx)
Rogelio F Jiménez-Ortega (frankunam_1@hotmail.com)
Alma Y Parra-Torres (yoli_379@hotmail.com)
Manuel Castillejos-López (mcastillejos@gmail.com)
Nelly Patiño (lnpatino@inmegen.gob.mx)
Manuel Quiterio-Trenado (mquitero@insp.mx)
Jorge Salmerón (jsalme@prodigy.net.mx)

Version: 2  Date: 20 August 2014

Author's response to reviews:

Manuscript Number: MS:1771369657140054

Title: Analysis of association of MEF2C, SOST and JAG1 genes with bone mineral density in Mexican-Mestizo postmenopausal women.

BMC Musculoskeletal Disorders

Response to Journal Editorial Office and list of changes made according to their comments.

Thank you for your comments; we hope you now find the manuscript acceptable.

Specific Points:

a) Title page: Please include email addresses of all authors at Title page.
   Answer: According to the Journal Editorial Office suggestion, are included email addresses of all authors in Title page.

b) Please revise your manuscript to include line and page numbers.
   Answer. According to the Journal Editorial Office suggestion, are included line and page numbers in the manuscript.

Sincerely,

Rafael Velázquez-Cruz, PhD
Laboratorio de Genómica del Metabolismo Óseo
Instituto Nacional de Medicina Genómica
Phone: +52 (55) 53501900 ext. 1149. Fax +52 (55) 53501999
E-mail: rvelazquez@inmegen.gob.mx